%~MoulinÃ© par MaN_auto v.0.33.4 2025-02-26 14:22:51
\documentclass[CRMATH,Unicode,XML,biblatex]{cedram}

\TopicFR{Physique mathÃ©matique, ProbabilitÃ©s}
\TopicEN{Mathematical physics, Probability theory}

\addbibresource{CRMATH_Mouzard_20240762.bib}

\usepackage{dsfont}
\usepackage{mathtools}

\newcommand{\IE}{\mathbb{E}}
\newcommand{\IN}{\mathbb{N}}
\newcommand{\IZ}{\mathbb{Z}}
\newcommand{\IR}{\mathbb{R}}
\newcommand{\IT}{\mathbb{T}}

\newcommand{\drm}{\mathrm d}

\newcommand{\CD}{\mathcal D}
\newcommand{\CS}{\mathcal S}
\newcommand{\CC}{\mathcal C}
\newcommand{\CH}{\mathcal H}
\newcommand{\CX}{\mathcal X}
\newcommand{\CB}{\mathcal B}
\newcommand{\SF}{\mathcal{F}}

\newcommand{\IDC}{\mathds{1}}

\renewcommand{\P}{\mathsf{P}}

\newcommand{\eps}{\varepsilon}

\DeclareMathOperator{\supp}{supp}

%%%---------------------------------------------------------------------------

%%% VERT_BAR

%%% ''Evaluated at'' macro
\newcommand\restr[2]{{% we make the whole thing an ordinary symbol
 \left.\kern-\nulldelimiterspace % automatically resize the bar with \right
 #1 % the function
 \vphantom{\big|} % pretend it's a little taller at normal size
 \right|_{#2} % this is the delimiter
 }}
% As a subscript: \phi_{F_f\rvert_{K_\ell}} [NO : \phi_{\left.h\right|_{K_\ell}}].

%%% \abs-like macros
%\usepackage{mathtools}
\DeclarePairedDelimiter{\abs}{\lvert}{\rvert}
\DeclarePairedDelimiter{\norm}{\lVert}{\rVert}
\newcommand{\normsub}{\norm*}					% Use with subscript.
% Star the operator for a scaling version.
% /!\ Attention: use case-sensitive Find&Replace.

%%% ''Such that'' macro
\newcommand*{\st}{\,\vert\,}
\newcommand*{\bigst}{\,\big\vert\,}
\newcommand*{\Bigst}{\,\Big\vert\,}
\newcommand*{\biggst}{\,\bigg\vert\,}
\newcommand*{\Biggst}{\,\Bigg\vert\,}
% Scaling version (with \left and \right delimiters): \,\middle\vert\,

%%% Divides
% \vert

%%% Separator (non-scaling)
% \mid

%%%---------------------------------------------------------------------------

%%% ARROWS

%%% \to macro
\renewcommand*{\to}{\mathchoice{\longrightarrow}{\rightarrow}{\rightarrow}{\rightarrow}}

%%% \mapsto macro
\let\oldmapsto\mapsto
\renewcommand*{\mapsto}{\mathchoice{\longmapsto}{\oldmapsto}{\oldmapsto}{\oldmapsto}}

%%% \longhookrightarrow macro
\newcommand{\longhookrightarrow}{\!\lhook\joinrel\longrightarrow}

%%%---------------------------------------------------------------------------

%%% "Better widebar" macro
\makeatletter
\let\save@mathaccent\mathaccent
\newcommand*\if@single[3]{%
  \setbox0\hbox{${\mathaccent"0362{#1}}^H$}%
  \setbox2\hbox{${\mathaccent"0362{\kern0pt#1}}^H$}%
  \ifdim\ht0=\ht2 #3\else #2\fi
  }
%The bar will be moved to the right by a half of \macc@kerna, which is computed by amsmath:
\newcommand*\rel@kern[1]{\kern#1\dimexpr\macc@kerna}
%If there's a superscript following the bar, then no negative kern may follow the bar;
%an additional {} makes sure that the superscript is high enough in this case:
\newcommand*\widebar[1]{\@ifnextchar^{{\wide@bar{#1}{0}}}{\wide@bar{#1}{1}}}
%Use a separate algorithm for single symbols:
\newcommand*\wide@bar[2]{\if@single{#1}{\wide@bar@{#1}{#2}{1}}{\wide@bar@{#1}{#2}{2}}}
\newcommand*\wide@bar@[3]{%
  \begingroup
  \def\mathaccent##1##2{%
%Enable nesting of accents:
    \let\mathaccent\save@mathaccent
%If there's more than a single symbol, use the first character instead (see below):
    \if#32 \let\macc@nucleus\first@char \fi
%Determine the italic correction:
    \setbox\z@\hbox{$\macc@style{\macc@nucleus}_{}$}%
    \setbox\tw@\hbox{$\macc@style{\macc@nucleus}{}_{}$}%
    \dimen@\wd\tw@
    \advance\dimen@-\wd\z@
%Now \dimen@ is the italic correction of the symbol.
    \divide\dimen@ 3
    \@tempdima\wd\tw@
    \advance\@tempdima-\scriptspace
%Now \@tempdima is the width of the symbol.
    \divide\@tempdima 10
    \advance\dimen@-\@tempdima
%Now \dimen@ = (italic correction / 3) - (Breite / 10)
    \ifdim\dimen@>\z@ \dimen@0pt\fi
%The bar will be shortened in the case \dimen@<0 !
    \rel@kern{0.6}\kern-\dimen@
    \if#31
      \overline{\rel@kern{-0.6}\kern\dimen@\macc@nucleus\rel@kern{0.4}\kern\dimen@}%
      \advance\dimen@0.4\dimexpr\macc@kerna
%Place the combined final kern (-\dimen@) if it is >0 or if a superscript follows:
      \let\final@kern#2%
      \ifdim\dimen@<\z@ \let\final@kern1\fi
      \if\final@kern1 \kern-\dimen@\fi
    \else
      \overline{\rel@kern{-0.6}\kern\dimen@#1}%
    \fi
  }%
  \macc@depth\@ne
  \let\math@bgroup\@empty \let\math@egroup\macc@set@skewchar
  \mathsurround\z@ \frozen@everymath{\mathgroup\macc@group\relax}%
  \macc@set@skewchar\relax
  \let\mathaccentV\macc@nested@a
%The following initialises \macc@kerna and calls \mathaccent:
  \if#31
    \macc@nested@a\relax111{#1}%
  \else
%If the argument consists of more than one symbol, and if the first token is
%a letter, use that letter for the computations:
    \def\gobble@till@marker##1\endmarker{}%
    \futurelet\first@char\gobble@till@marker#1\endmarker
    \ifcat\noexpand\first@char A\else
      \def\first@char{}%
    \fi
    \macc@nested@a\relax111{\first@char}%
  \fi
  \endgroup
}
\makeatother

\let\oldbar\bar
\renewcommand*{\bar}[1]{{\mathchoice{\widebar{#1}}{\widebar{#1}}{\widebar{#1}}{\oldbar{#1}}}}

% /!\ Il y a parfois des soucis avec cette macro lorsqu'il y a des caractÃ¨res en minuscule en subscript (et dans les captions). Dans ce cas, faire avec \overline et des espaces manuels.

%%% Double bar
% \,\overline{\overline{\!a}}=a

%%%---------------------------------------------------------------------------

\newcommand\bx{\,\overline{\!x}}
\newcommand\brho{\,\overline{\!\rho}}
%better widebar a apparemment des soucis avec les \subcaptions pour les CR :shrug:

%%%---------------------------------------------------------------------------

\graphicspath{{./figures/}}

\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}

\let\oldtilde\tilde
\renewcommand*{\tilde}[1]{\mathchoice{\widetilde{#1}}{\widetilde{#1}}{\oldtilde{#1}}{\oldtilde{#1}}}
\let\oldhat\hat
\renewcommand*{\hat}[1]{\mathchoice{\widehat{#1}}{\widehat{#1}}{\oldhat{#1}}{\oldhat{#1}}}
\let\oldexists\exists
\renewcommand*{\exists}{\mathrel{\oldexists}}
\let\oldforall\forall
\renewcommand*{\forall}{\mathrel{\oldforall}}

\title{A simple construction of the Anderson operator via its quadratic form in dimensions two and three}

\alttitle{Une construction simple de l'opÃ©rateur d'Anderson par sa forme quadratique en dimensions deux et trois}

\author{\firstname{Antoine} \lastname{Mouzard}}
\address{Modal'X -- UMR CNRS 9023, UniversitÃ© Paris Nanterre, 92000 Nanterre, France}
\email{antoine.mouzard@math.cnrs.fr}

\author{\firstname{El Maati} \lastname{Ouhabaz}}
\address{UniversitÃ© de Bordeaux, IMB, 351 cours de la LibÃ©ration, 33405 Talence, France}
\email{elmaati.ouhabaz@math.u-bordeaux.fr}

\subjclass{60H25, 60H17, 35J10}

\keywords{\kwd{Anderson form} \kwd{singular stochastic operator} \kwd{SchrÃ¶dinger operator} \kwd{renormalization} \kwd{positivity} \kwd{spectral gap}}
\altkeywords{\kwd{Forme d'Anderson} \kwd{opÃ©rateurs stochastiques singuliers} \kwd{opÃ©rateur de SchrÃ¶dinger} \kwd{renormalisation} \kwd{positivitÃ©} \kwd{trou spectral}}

\begin{abstract}
We provide a simple construction of the Anderson operator in dimensions two and three. This is done through its quadratic form. We rely on an exponential transform instead of the regularity structures or paracontrolled calculus which are usually used for the construction of the operator. The knowledge of the form is robust enough to deduce important properties such as positivity and irreducibility of the corresponding semigroup. The latter property gives existence of a spectral gap.
\end{abstract}

\begin{altabstract}
Nous fournissons une construction simple de l'opÃ©rateur d'Anderson en dimensions deux et trois. Cela est rÃ©alisÃ© Ã  travers sa forme quadratique. Nous nous appuyons sur une transformation exponentielle au lieu des structures de rÃ©gularitÃ© ou du calcul paracontrÃ´lÃ©, qui sont gÃ©nÃ©ralement utilisÃ©s pour la construction de l'opÃ©rateur. La connaissance de la forme est suffisamment robuste pour dÃ©duire des propriÃ©tÃ©s importantes telles que la positivitÃ© et l'irrÃ©ductibilitÃ© du semi-groupe correspondant. Cette derniÃ¨re propriÃ©tÃ© permet de dÃ©montrer l'existence d'un trou spectral.
\end{altabstract}

\begin{document}

\maketitle

\section*{Introduction}

Over the last decade, the study of singular stochastic PDEs has grown to an important field with the introduction of regularity structures by Hairer~\cite{Hai14} and paracontrolled calculus by Gubinelli, Imkeller and Perkowski~\cite{GIP}. The theory first aimed at the resolution of parabolic equations such as the Parabolic Anderson Model (PAM) equation or the Kardar--Parisi--Zhang (KPZ) equation, it then led to the construction of the Anderson Hamiltonian
\begin{equation*}
H=-\Delta+\xi
\end{equation*}
with $\xi$ the spatial white noise, see~\cite{AllezChouk,GUZ,Labbe,CVZ,Mouzard} in dimension~$2$ and $3$, on a finite box with periodic or Dirichlet boundary conditions or even compact Riemannian manifolds.

In this note we provide a simple construction of this operator via its quadratic form without using regularity structures or paracontrolled calculus. We rely on an exponential transform first used by Hairer and LabbÃ©~\cite{HairerLabbe15} for the continuum parabolic Anderson model on $\IR^2$ and then used in different context, see for example~\cite{DW,TV,JagannathPerkowski23,BDFT23one,ChauleurMouzard23}. In particular, this was already used by Matsuda and van Zuijlen~\cite{MatsudaZuijlen22} to construct the rough form in the full subcritical regime using also regularity structures. Our work can be seen as an unsophisticated approach since no regularity structures or paracontrolled calculus appear here. See also~\cite{MM,Mouzard22} for other singular stochastic operators.

The Anderson Hamiltonian is the SchrÃ¶dinger operator $H=-\Delta+\xi$ with $\xi$ the space white noise which is a random distribution of negative HÃ¶lder regularity $-\frac{d}{2}-\kappa$ for any $\kappa>0$. In one dimension, it is the derivative of the Brownian motion and the associated form
\begin{equation*}
\mathfrak{a}(u,v)=\int_0^1\nabla u(x)\cdot\nabla v(x)\drm x+\int_0^1u(x)v(x)\xi(\drm x)
\end{equation*}
was constructed by Fukushima and Nakao~\cite{FN} with domain the usual Sobolev space $\CH^1$. The idea is that one can multiply two distributions if the sum of their regularity is positive, hence $uv\xi$ is well-defined as a distribution for $u,v\in\CH^1$ since $\xi\in\CC^{-\frac{1}{2}-\kappa}$. In two dimensions, $\xi\in\CC^{-1-\kappa}$ and this construction is not possible anymore. Following recent progress on singular stochastic PDEs, the operator can be constructed with a random domain $\CD_\xi^2$ depending on the noise $\xi$ such that $H\colon\CD_\xi^2\subset L^2\to L^2$ is an unbounded closed operator. Taking $u\in L^2$ and assuming that $Hu$ is an element of $L^2$, one obtains the relation
\begin{equation*}
\Delta u=u\xi-Hu
\end{equation*}
which induces an expansion of $u$ with respect to the noise using regularity structures or paracontrolled calculus. In particular, \cite{GUZ} and~\cite{Mouzard} also identify a form domain, that is a random subspace $\CD_\xi^1\subset L^2$ such that
\begin{equation*}
\forall u\in\CD_\xi^1,\quad\bigl\lvert\langle Hu,u\rangle\bigr\rvert<\infty.
\end{equation*}
We emphasize in the notation that the domains are random and consist of random functions depending on the noise. In each case, the operator is a singular stochastic operator and a renormalization procedure is involved in its construction. For a regularization $\xi_\eps$ of the noise, the operator is constructed as a limit in the resolvent sense, that is
\begin{equation*}
H=\lim_{\eps\to0}\bigl(-\Delta+\xi_\eps-c_\eps\bigr)
\end{equation*}
with a constant $c_\eps$ which explodes when $\eps \to 0$. It is related to the definition of the product $\Delta^{-1}\xi\cdot\xi$ and the divergence of the Green function of the Laplacian. In two dimensions, one has $c_\eps\sim\log(\eps)$ while $c_\eps\sim\eps^{-1}$ in three dimensions. In one dimension, this product is well-defined and one can take $c_\eps=0$ which is coherent with~\cite{FN} and $\CD_\xi^1=\CH^1$ does not depend on the noise. However the domain of the operator is random and this method was recently used by Dumaz and LabbÃ© to provide a precise study of the operator, see~\cite{DumazLabbe22} and references therein. The Anderson form also appears as the energy for dispersive PDEs such as the nonlinear SchrÃ¶dinger equation
\begin{equation*}
i\partial_tu=\Delta u+u\xi+\abs{u}^2u
\end{equation*}
and was used to obtain solutions, see for example~\cite{DW,GUZ,Mouzard,TV,MZ,ChauleurMouzard23} and references therein. In particular, uniform bounds in energy is the crucial property of such singular stochastic PDEs where one does not have the regularizing properties of the parabolic equation. In this context, one has to work with random initial data depending on the noise and the conservation of energy makes the form domain of the Anderson operator a natural space to get a global solution.

In this work, we consider a new variable $u=e^Xv$ for a suitable random field $X$. In this case, we have
\begin{equation*}
\Delta u=e^X\Delta v+2e^X\nabla X\cdot\nabla v+e^X\bigl(\abs{\nabla X}^2+\Delta X\bigr)v
\end{equation*}
and if $X$ is a solution to $\Delta X=\xi$, the Anderson operator is formaly given by
\begin{equation*}
Hu=-e^X\Delta v-2e^X\nabla X\cdot\nabla v-e^X\abs{\nabla X}^2v.
\end{equation*}
In two dimensions, we have
\begin{equation*}
\xi\in\CC^{-1-\kappa}\quad\implies\quad\nabla X\in\CC^{-\kappa},
\end{equation*}
hence the square $\abs{\nabla X}^2$ is singular and has to be defined with a renormalization procedure as a Wick product $\abs{\nabla X}^{2\diamond}\in\CC^{-2\kappa}$. In this case, $v\in\CH^1$ is regular enough for the associated form to make sense and one can construct the Anderson form with domain $\CD(\mathfrak{a})=e^X\CH^1$. In three dimensions, we have
\begin{equation*}
\xi\in\CC^{-\frac{3}{2}-\kappa}\quad\implies\quad\nabla X\in\CC^{-\frac{1}{2}-\kappa}
\end{equation*}
and the Wick product $\abs{\nabla X}^{2\diamond}\in\CC^{-1-2\kappa}$ is too rough to be multiplied by $v\in\CH^1$. One can apply the same method and construct the Anderson form with domain $e^{X+Y}\CH^1$ with a suitable second random field $Y$.

This exponential transform allows us to construct a symmetric form $\mathfrak{a}$ whose associated operator $H$ is the Anderson Hamiltonian. Once $\mathfrak{a}$ is defined, the associated operator is given~by
\begin{equation*}
\CD(H)=\bigl\{u\in L^2(\IT^d)\ ;\ \exists v\in L^2(\IT^d),\forall\varphi\in\CD(\mathfrak{a}), \mathfrak{a}(u,\varphi)=\langle v,\varphi\rangle\bigr\}, \quad H u = v.
\end{equation*}
This operator is well defined with dense domain $\CD(H)$ in $L^2(\IT^d)$. This weak formulation does not however give a precise description of $\CD(H)$. Also, the approach using the exponential transform does not allow to construct the explicit domain of the operator. For this a more involved theory such as regularity structures or paracontrolled calculus seems to be necessary. See Section~\ref{SectionSpectralGap} for some additional details. Nevertheless, the knowledge of $H$ through its form is enough to deduce that $H$ is self-adjoint, it has a discrete spectrum and an $L^2$ orthonormal basis given by eigenfunctions. In addition, relying on a criterion from~\cite{Ouhabaz05}, we prove that the associated semigroup is irreducible. In particular, this implies the existence of a spectral gap $\lambda_1<\lambda_2$ with a positive ground state $\Psi\in\CD(H)$. This result was already proved in~\cite{BailleulDangMouzard22} by relying on a quantitative estimate for the linear Parabolic Anderson Model equation. Our work provides a pedestrian approach to this result even in three dimensions which usually relies on involved computations with expansion of order $5$ using regularity structures or paracontrolled calculus.

In order to keep the ideas and tools simple we restrict ourselves to the case of the torus $\IT^d$ for $d \in \{2,3\}$ (endowed with the Lebesgue measure $\drm x$). Our construction works on any compact manifold without boundary even though we rely on Fourier series here. On a manifold, one can use CalderÃ³n--Zygmund formula as in~\cite[Section~2.1]{Mouzard} with tools based on the heat semigroup of the Laplace--Beltrami operator on the manifold. Similar renormalization problems were also considered in~\cite{BailleulDangMouzard22} with microlocal and harmonic analysis on local charts.

In the work~\cite{MatsudaZuijlen22}, Matsuda and van Zuijlen use the exponential transform to construct the Dirichlet form associated to a SchrÃ¶dinger operator for a general class of random potentials in the full subcritical regime in the sense of singular SPDEs using regularity structures, that is dimension $d\in\{2,3\}$ as far as the Anderson hamiltonian is concerned. To do so, they use higher order iteration of the argument we use in this note and for which an arbitrary large number of singular stochastic terms need to be renormalized. They rely on the renormalization algorithm from the very involved works~\cite{CH,BHZ} which form two important part of the full theory of regularity structures with~\cite{Hai14,BCCH18}. In particular, they show that the exponential transform can be used in the full subcritical regime. The present work aims at making accessible such interesting ideas without the knowledge of regularity structures, we only rely on a direct renormalization and Fourier series while still covering the case of the Anderson Hamiltonian in two and three dimensions.

In Section~\ref{SectionStoc}, we provide several bounds on stochastic functions and distributions that we need to construct the form. In particular, this is where the renormalization of the singular products is done. In Sections~\ref{Section2D} and~\ref{Section3D}, we respectively construct the Anderson form in two and three dimensions using the first and second order exponential transform. In Section~\ref{SectionSpectralGap}, we prove irreducibility and existence of a spectral gap.

\section{Stochastic bounds and renormalization} \label{SectionStoc}

On the torus $\IT^d$, the white noise is given by
\begin{equation*}
\xi(x)=\sum_{k\in\IZ^d}\xi_ke^{ik\cdot x}
\end{equation*}
with $(\xi_k)_{k\in\IZ^d}$ a family of independent and identically distributed random variables of centered standard complex Gaussian with $\xi_{-k}=\bar{\xi_k}$. This gives a centered real Gaussian field with covariance function
\begin{equation*}
\IE\bigl[\xi(x)\xi(y)\bigr]=\delta_0(x-y),
\end{equation*}
that is a random distribution $(\langle\xi,\varphi\rangle)_{\varphi\in L^2(\IT^d)}$ such that
\begin{equation*}
\IE\bigl[\langle\xi,\varphi\rangle\langle\xi,\psi\rangle\bigr]=\langle\varphi,\psi\rangle_{L^2(\IT^d)}
\end{equation*}
which indeed gives the Fourier coefficient $(\xi_k)_{k\in\IZ^d}$. Because of the lack of decay at infinity of the Fourier coefficients, this sum has to be interpreted in a weak sense. Its first construction is due to Paley and Zygmund~\cite{PaleyZygmund30,PaleyZygmund32} and is actually the first random distribution ever considered. A natural and convenient setting is given by the Besov space $\CB_{p,q}^\alpha$ which can be defined using the Littlewood--Paley decomposition, see for example~\cite{BCD}. This decomposition can be stated as follows
\begin{equation*}
u=\sum_{n\ge0}\Delta_nu
\end{equation*}
with $\Delta_nu=\bigl(\SF^{-1}\IDC_{\abs{\,\cdot\,}\simeq 2^n}\SF\bigr)u$, that is the projection of $u$ in frequencies on an annulus of size $2^n$. It is defined by
\begin{equation*}
\bigl(\Delta_nu\bigr)(x)\coloneqq 2^{d(n-1)}\int_{\IR^d}\chi\bigl(2^{n-1}(x-y)\bigr)u(y)\drm y
\end{equation*}
with $\chi\in\CS(\IR^d)$ and $\supp\widehat\chi\subset\bigl\{\frac{1}{2}\le\abs{z}\le 2\bigr\}$ for $n\ge1$ and
\begin{equation*}
\bigl(\Delta_0u\bigr)(x)\coloneqq \int_{\IR^d}\chi_0(x-y)u(y)\drm y
\end{equation*}
with $\chi_0\in\CS(\IR^d)$ and $\supp\widehat\chi_0\subset\bigl\{\abs{z}\le 1\bigr\}$. We also denote $K=\widehat{\chi}$ such that $\Delta_nu=\bigl(\SF^{-1}K(2^n\,\cdot\,)\SF\bigr)u$. Then the Besov space $\CB_{p,q}^\alpha$ are distributions such that
\begin{equation*}
\norm{u}_{\CB_{p,q}^\alpha}\coloneqq \Biggl( \sum_{n\ge0}2^{\alpha pn}\norm{\Delta_nu}_{L^q(\IT^d)}^p \Biggr)^{\frac{1}{p}}<\infty.
\end{equation*}
The particular case $p=q=2$ corresponds to the Sobolev space $\CB_{2,2}^\alpha=\CH^\alpha$ and for $p=q=\infty$ with $\alpha\in\IR_+\backslash\IN$, one gets the usual HÃ¶lder spaces $\CB_{\infty,\infty}^\alpha=\CC^\alpha$. One also has the continuous Besov embedding
\begin{equation*}
\CB_{p_1,q_1}^\alpha\longhookrightarrow \CB_{p_2,q_2}^{\alpha-d\bigl(\frac{1}{p_1}-\frac{1}{p_2}\bigr)}
\end{equation*}
for $p_1\le p_2$, $q_1\le q_2$ and $\alpha\in\IR$. While one can a priori only multiply a distribution by a smooth function, one has the following product rule in the case of Besov spaces which corresponds to Young condition.

\begin{proposition}\label{PropProduct}
For $\alpha,\beta\in\IR$ such that $\alpha+\beta>0$ and $p,q,r\in[1,\infty]$ such that $\frac{1}{r}=\frac{1}{p}+\frac{1}{q}$, there exists a constant $C>0$ such that
\begin{equation*}
\norm{uv}_{\CB_{r,r}^{\alpha\wedge\beta}}\le C\norm{u}_{\CB_{p,p}^\alpha}\norm{v}_{\CB_{q,q}^\beta}.
\end{equation*}
\end{proposition}

The following proposition gives a similar result at the level of the duality bracket.

\begin{proposition}\label{PropDuality}
For $\alpha\in\IR$ and $p,p',q,q'\in[1,\infty]$ such that $1=\frac{1}{p}+\frac{1}{p'}=\frac{1}{q}+\frac{1}{q'}$, there exists a constant $C>0$ such that
\begin{equation*}
\bigl\lvert\langle u,v\rangle\bigr\rvert\le C\norm{u}_{\CB_{p,q}^\alpha}\norm{v}_{\CB_{p',q'}^{-\alpha}}.
\end{equation*}
\end{proposition}

For later use, we introduce a new random field $X$ defined by
\begin{equation*}
X(x)=\sum_{k\in\IZ^d\backslash\{0\}}\frac{1}{\abs{k}^2}\xi_ke^{ik\cdot x}.
\end{equation*}
It satisfies the equation
\begin{equation*}
\Delta X=\xi-\xi_0.
\end{equation*}

The following proposition gives HÃ¶lder regularity of $\xi$ and $X$. This result is well-known but we give a proof for self-completeness and also to give a flavor of the arguments for the stochastic renormalization.

\begin{proposition}
For any\/ $\kappa>0$, one has almost surely
\begin{equation*}
\xi\in\CC^{-\frac{d}{2}-\kappa}(\IT^d)\quad\text{ and }\quad X\in\CC^{2-\frac{d}{2}-\kappa}(\IT^d).
\end{equation*}
\end{proposition}

\begin{proof}
Since the noise is Gaussian, we have
\begin{equation*}
\IE\bigl[\langle \xi,\varphi\rangle^p\bigr]\le(p-1)^{\frac{p}{2}}\IE\bigl[\langle\xi,\varphi\rangle^2\bigr]^{\frac{p}{2}}
\end{equation*}
for any test function $\varphi$. This is usually referred to as Gaussian hypercontractivity. In order to use this, we estimate the Besov norm $\CB_{p,p}^\gamma$ for $p$ large and use the embedding
\begin{equation*}
\CB_{p,p}^\gamma(\IT^d) \longhookrightarrow \CB^{\gamma-\frac{d}{p}}_{\infty,\infty}(\IT^d).
\end{equation*}
We have
\begin{align*}
\IE\Bigl[\norm{\Delta_n\xi}_{L^p(\IT^d)}^p\Bigr]&=\int_{\IT^d}\IE\bigl[\bigl\langle \xi,\chi_n(x-\,\cdot\,)\bigr\rangle^p\bigr]\drm x\\
&\le(p-1)^{\frac{p}{2}}\int_{\IT^d}\IE\bigl[\bigl\langle \xi,\chi_n(x-\,\cdot\,)\bigr\rangle^2\bigr]^{\frac{p}{2}}\drm x\\
&\le(p-1)^{\frac{p}{2}}\norm{\chi_n}_{L^2(\IT^d)}^p\abs{\IT^d}
\end{align*}
with $\chi_n(\,\cdot\,)=2^{dn}\chi(2^n\,\cdot\,)$ and using that $\xi$ is an isometry from $L^2(\IT^d)$ to $L^2(\Omega)$. We have
\begin{equation*}
\norm{\chi_n}_{L^2(\IT^d)}^2=2^{2dn}\norm{\chi(2^n\,\cdot\,)}_{L^2(\IT^d)}^2=2^{dn}\norm{\chi}_{L^2(\IT^d)}^2
\end{equation*}
hence
\begin{equation*}
\IE\Bigl[\norm{\Delta_n\xi}_{L^p(\IT^d)}^p\Bigr]\le(p-1)^{\frac{p}{2}}2^{pn\frac{d}{2}}\norm{\chi}_{L^2(\IT^d)}^p\abs{\IT^d}.
\end{equation*}
This gives
\begin{equation*}
\IE\Bigl[\norm{\xi}_{\CB_{p,p}^{-\frac{d}{2}}}\Bigr]<\infty,
\end{equation*}
and hence $\xi\in\CC^{-\frac{d}{2}-\frac{d}{p}}(\IT^d)$ for any $p\ge1$ which completes the proof for the regularity of $\xi$ while the regularity of $X$ follows from a standard regularity estimate.
\end{proof}

In two dimensions, one has $X\in\CC^{1-\kappa}$ hence $\nabla X\in\CC^{-\kappa}$ and the square $\abs{\nabla X}^2$ is ill-defined since $-2\kappa<0$. Consider a regularization of the noise $\xi_\eps=\xi*\rho_\eps$ with $\rho_\eps$ a mollifier, radial to simplify the computations. Then $\xi_\eps$ converges to $\xi$ as $\eps$ goes to $0$ in $\CC^{-1-\kappa}$ and one can consider $X_\eps$ the solution with null mean to
\begin{equation*}
\Delta X_\eps=\xi_\eps-\langle\xi_\eps,1\rangle
\end{equation*}
which converges to $X$ in $\CC^{1-\kappa}$ as $\eps$ goes to $0$. Since the square $\abs{\nabla X}^2$ is ill-defined, the quantity $\abs{\nabla X_\eps}^2$ diverges and this is described by the Wick square as proved in the following proposition.

\begin{proposition}
There exists a distribution $\abs{\nabla X}^{2\diamond}\in\CC^{-2\kappa}(\IT^2)$ such that
\begin{equation*}
\abs{\nabla X}^{2\diamond}=\lim_{\eps\to0}\Bigl(\abs{\nabla X_\eps}^2-\IE\bigl[\abs{\nabla X_\eps}^2\bigr]\Bigr)
\end{equation*}
in $\CC^{-2\kappa}(\IT^2)$ in probability. Moreover, one has
\begin{equation*}
\IE\bigl[\abs{\nabla X_\eps}^2\bigr]\underset{\eps\to0}{\sim}-\frac{1}{(2\pi)^2}\log(\eps).
\end{equation*}
\end{proposition}

\begin{proof}
Since $\xi_\eps=\xi*\rho_\eps$, we have
\begin{equation*}
X_\eps(x)=\sum_{k\in\IZ^2\backslash\{0\}}\frac{\widehat{\rho_\eps}(k)}{\abs{k}^2}\xi_ke^{ikx}
\end{equation*}
thus
\begin{equation*}
\bigl\lvert\nabla X_\eps(x)\bigr\rvert^2=\sum_{k,k'\in\IZ^2\backslash\{0\}}\theta\bigl(\eps\abs{k}\bigr)\theta\bigl(\eps\abs{k'}\bigr)\frac{k\cdot k'}{\abs{k}^2\abs{k'}^2}\xi_k\bar{\xi_{k'}}e^{i(k-k')\cdot x}
\end{equation*}
with $\theta\bigl(\eps\abs{k}\bigr)\coloneqq \widehat{\rho_\eps}(k)$ since $\rho$ is radial. Using $\IE\bigl[\xi_k\bar{\xi_{k'}}\ \bigr]=\delta_0(k-k')$, we have
\begin{equation*}
\IE\Bigl[\bigl\lvert\nabla X_\eps(x)\bigr\rvert^2\Bigr]=\sum_{k\in\IZ^2\backslash\{0\}}\frac{\theta\bigl(\eps\abs{k}\bigr)^2}{\abs{k}^2}\underset{\eps\to0}{\sim}\frac{1}{(2\pi)^2}\log(\eps)
\end{equation*}
which gives the second part of the statement. For $n\ge1$, we have
\begin{equation*}
\Delta_n\Bigl(\abs{\nabla X_\eps}^2-\IE\bigl[\abs{\nabla X_\eps}^2\bigr]\Bigr)(x)=\sum_{k\neq k'}K_n(k-k')\theta\bigl(\eps\abs{k}\bigr)\theta\bigl(\eps\abs{k'}\bigr)\frac{k\cdot k'}{\abs{k}^2\abs{k'}^2}\xi_k\bar{\xi_{k'}}e^{i(k-k')\cdot x}
\end{equation*}
with $K_n(\,\cdot\,)=K(2^{-n}\,\cdot\,)$ since $\Delta_n\IE\bigl[\abs{\nabla X_\eps}^2\bigr]=0$ for $n\ge1$, the expectation of $\abs{\nabla X_\eps}^2$ being a constant function. Then the expectation of $\Bigl\lvert\Delta_n\Bigl(\abs{\nabla X_\eps}^2-\IE\bigl[\abs{\nabla X_\eps}^2\bigr]\Bigr)\Bigr\rvert^2(x)$ is given by
\begin{multline*}
\sum_{\substack{k_1\neq k_1'\\k_2\neq k_2'}}K_n(k_1-k_1')\bar{K_n(k_2-k_2')}\theta\bigl(\eps\abs{k_1}\bigr)\theta\bigl(\eps\abs{k_1'}\bigr)\theta\bigl(\eps\abs{k_2}\bigr)\theta\bigl(\eps\abs{k_2'}\bigr)\frac{k_1\cdot k_1'}{\abs{k_1}^2\abs{k_1'}^2}\frac{k_2\cdot k_2'}{\abs{k_2}^2\abs{k_2'}^2}\IE\bigl[\xi_{k_1}\bar{\xi_{k_1'}}\xi_{k_2}\bar{\xi_{k_2'}}\bigr]\\
\times e^{i(k_1-k_1')\cdot x}e^{-i(k_2-k_2')\cdot x}
\end{multline*}
and we have
\begin{align*}
\IE\bigl[\xi_{k_1}\bar{\xi_{k_1'}}\xi_{k_2}\bar{\xi_{k_2'}}\bigr]&=\IE\bigl[\xi_{k_1}\bar{\xi_{k_1'}}\bigr]\IE\bigl[\xi_{k_2}\bar{\xi_{k_2'}}\bigr]+\IE\bigl[\xi_{k_1}\xi_{k_2}\bigr]\IE\bigl[\bar{\xi_{k_1'}\xi_{k_2'}}\bigr]+\IE\bigl[\xi_{k_1}\bar{\xi_{k_2'}}\bigr]\IE\bigl[\bar{\xi_{k_1'}}\xi_{k_2}\bigr]\\
&=\delta_0(k_1-k_1')\delta_0(k_2-k_2')+\delta_0(k_1+k_2)\delta_0(k_1'+k_2')+\delta_0(k_1-k_2')\delta_0(k_1'-k_2)
\end{align*}
for any $k_1,k_1',k_2,k_2'\in\IZ^2$. It follows that
\begin{align*}
\IE\Bigl\lvert\Delta_n\Bigl(\abs{\nabla X_\eps}^2-\IE\bigl[\abs{\nabla X_\eps}^2\bigr]\Bigr)\Bigr\rvert^2&=2\sum_{k_1,k_2}\bigl\lvert K_n(k_1-k_2)\bigr\rvert^2\frac{\theta\bigl(\eps\abs{k_1}\bigr)^2\theta\bigl(\eps\abs{k_2}\bigr)^2}{\abs{k_1}^2\abs{k_2}^2}\\
&=2\sum_{k_1,k_2}\bigl\lvert K\bigl(2^{-n}(k_1-k_2)\bigr)\bigr\rvert^2\frac{\theta\bigl(\eps\abs{k_1}\bigr)^2\theta\bigl(\eps\abs{k_2}\bigr)^2}{\abs{k_1}^2\abs{k_2}^2}\\
&=2\sum_k\bigl\lvert K(2^{-n}k)\bigr\rvert^2\sum_{k_1-k_2=k}\frac{\theta\bigl(\eps\abs{k_1}\bigr)^2\theta\bigl(\eps\abs{k_2}\bigr)^2}{\abs{k_1}^2\abs{k_2}^2}\\
&=2\sum_k\bigl\lvert K(2^{-n}k)\bigr\rvert^2\sum_{k_2}\frac{\theta(\eps\abs{k+k_2})^2\theta\bigl(\eps\abs{k_2}\bigr)^2}{\abs{k+k_2}^2\abs{k_2}^2}\\
&\le C2^{2n}2^{-(2-2\kappa)n}\sum_{k_2}\frac{\theta\bigl(\eps\abs{k_2}\bigr)^2}{\abs{k_2}^{2+2\kappa}}
\end{align*}
for any $\kappa>0$ and a constant $C>0$ using the support of $K$. The Gaussian hypercontractivity yields
\begin{align*}
\IE\Bigl\lvert\Delta_n\Bigl(\abs{\nabla X_\eps}^2-\IE\bigl[\abs{\nabla X_\eps}^2\bigr]\Bigr)\Bigr\rvert^p&\le(p-1)^p\Biggl(\IE\biggl\lvert\Delta_n\Bigl(\bigl\lvert\nabla X_\eps(x)\bigr\rvert^2-\IE\Bigl[\bigl\lvert\nabla X_\eps(x)\bigr\rvert^2\Bigr]\Bigr)\biggr\rvert^2\Biggr)^{\frac{p}{2}}\\
&\le C2^{\kappa np}.
\end{align*}
Thus, $\abs{\nabla X_\eps}^2-\IE\bigl[\abs{\nabla X_\eps}^2\bigr]$ is bounded in $\CB_{p,p}^{-\kappa}$ for any $\kappa>0$ and $p\ge1$. Using the embedding $\CB_{p,p}^{-\kappa}\hookrightarrow\CC^{-\kappa-\frac{d}{p}}$ and a similar bound, one proves that $\bigl(\abs{\nabla X_\eps}^2-\IE\bigl[\abs{\nabla X_\eps}^2\bigr]\bigr)_{\eps>0}$ is a Cauchy family in $\CC^{-\kappa}$ for any $\kappa>0$ which completes the proof.
\end{proof}

We define the two dimensional enhanced noise
\begin{equation*}
\Xi=\bigl(\xi,\abs{\nabla X}^{2\diamond}\bigr)
\end{equation*}
which belongs to
\begin{equation*}
\CX^\kappa(\IT^2)=\CC^{-1-\kappa}(\IT^2)\times\CC^{-2\kappa}(\IT^2)
\end{equation*}
for any $\kappa>0$. We also have that
\begin{equation*}
\Xi_\eps=\bigl(\xi_\eps,\abs{\nabla X_\eps}^2-(2\pi)^{-2}\log(\eps)\bigr)
\end{equation*}
converges to $\Xi$ in $\CX^\kappa(\IT^2)$ for any $\kappa>0$. In three dimensions, one has $X\in\CC^{\frac{1}{2}-\kappa}$ hence this term is even more singular with $-1-2\kappa<0$. The analog of the previous renormalization is the following proposition with a larger divergence. Its proof follows the same path as the previous one.
\begin{proposition}
There exists a distribution $\abs{\nabla X}^{2\diamond}\in\CC^{-1-2\kappa}(\IT^3)$ such that
\begin{equation*}
\abs{\nabla X}^{2\diamond}=\lim_{\eps\to0}\Bigl(\abs{\nabla X_\eps}^2-\IE\bigl[\abs{\nabla X_\eps}^2\bigr]\Bigr)
\end{equation*}
in $\CC^{-1-2\kappa}(\IT^3)$ in probability. Moreover, one has
\begin{equation*}
\IE\bigl[\abs{\nabla X_\eps}^2\bigr]\underset{\eps\to0}{\sim}-\frac{1}{(2\pi)^2}\frac{1}{\eps}.
\end{equation*}
\end{proposition}

Since the noise is more irregular, $\abs{\nabla X}^{2\diamond}$ is too rough to make sense of its bracket with $\CH^1$ hence we will also need $Y$ the solution to
\begin{equation*}
\Delta Y=\abs{\nabla X}^{2\diamond}-\bigl\langle\abs{\nabla X}^{2\diamond},1\bigr\rangle
\end{equation*}
which belongs to $\CC^{1-2\kappa}(\IT^3)$. The square of its gradient is also singular and can be defined as a Wick product, as well as the product $\nabla X\cdot\nabla Y$. Note that for this product, there is no renormalization constant due to an algebraic cancellation.

\begin{proposition}
There exists a distribution $\abs{\nabla Y}^{2\diamond}\in\CC^{-4\kappa}(\IT^3)$ such that
\begin{equation*}
\abs{\nabla Y}^{2\diamond}=\lim_{\eps\to0}\Bigl(\abs{\nabla Y_\eps}^2-\IE\bigl[\abs{\nabla Y_\eps}^2\bigr]\Bigr)
\end{equation*}
in $\CC^{-4\kappa}(\IT^3)$ in probability. Moreover, one has
\begin{equation*}
\IE\bigl[\abs{\nabla Y_\eps}^2\bigr]\underset{\eps\to0}{\sim}-C\log(\eps)
\end{equation*}
with $C>0$ a constant. There also exists a distribution $\nabla X\diamond\nabla Y\in\CC^{-\frac{1}{2}-3\kappa}(\IT^3)$ such that
\begin{equation*}
\nabla X\diamond\nabla Y=\lim_{\eps\to0}\bigl(\nabla X_\eps\cdot\nabla Y_\eps\bigr).
\end{equation*}
\end{proposition}

\begin{proof}
We have
\begin{equation*}
Y_\eps(x)=\sum_{\substack{k,k'\neq 0\\k\neq k'}}\theta\bigl(\eps\abs{k}\bigr)\theta\bigl(\eps\abs{k'}\bigr)\frac{k\cdot k'}{\abs{k}^2\abs{k'}^2}\xi_k\bar{\xi_{k'}}\frac{e^{i(k-k')\cdot x}}{\abs{k-k'}^2}
\end{equation*}
hence
\begin{align*}
\bigl\lvert\nabla Y_\eps(x)\bigr\rvert^2&=\sum_{\substack{k_1\neq k_1'\\k_2\neq k_2'\\k_1,k_1',k_2,k_2'\neq 0}}\theta\bigl(\eps\abs{k_1}\bigr)\theta\bigl(\eps\abs{k_1'}\bigr)\theta\bigl(\eps\abs{k_2}\bigr)\theta\bigl(\eps\abs{k_2'}\bigr)\frac{k_1\cdot k_1'}{\abs{k_1}^2\abs{k_1'}^2}\frac{k_2\cdot k_2'}{\abs{k_2}^2\abs{k_2'}^2}\xi_{k_1}\bar{\xi_{k_1'}}\xi_{k_2'}\bar{\xi_{k_2}}\\
&\qquad\qquad\frac{(k_1-k_1')\cdot(k_2-k_2')}{\abs{k_1-k_1'}^2\abs{k_2-k_2'}^2}e^{i(k_1-k_1')\cdot x}e^{i(k_2'-k_2)\cdot x}\\
&=\sum_{n\in\IZ^3}e^{in\cdot x}\widehat{\abs{\nabla Y_\eps}^2}(n)
\end{align*}
with $\widehat{\abs{\nabla Y_\eps}^2}(n)$ given by
\begin{equation*}
\sum_{\substack{k_1-k_1'+k_2'-k_2=n\\k_1\neq k_1',k_2\neq k_2'\\k_1,k_1',k_2,k_2'\neq 0}}\theta\bigl(\eps\abs{k_1}\bigr)\theta\bigl(\eps\abs{k_1'}\bigr)\theta\bigl(\eps\abs{k_2}\bigr)\theta\bigl(\eps\abs{k_2'}\bigr)\xi_{k_1}\bar{\xi_{k_1'}}\xi_{k_2'}\bar{\xi_{k_2}}\frac{k_1\cdot k_1'}{\abs{k_1}^2\abs{k_1'}^2}\frac{k_2\cdot k_2'}{\abs{k_2}^2\abs{k_2'}^2}\frac{(k_1-k_1')\cdot(k_2'-k_2)}{\abs{k_1-k_1'}^2\abs{k_2-k_2'}^2}.
\end{equation*}
Using again
\begin{equation*}
\IE\bigl[\xi_{k_1}\bar{\xi_{k_1'}}\xi_{k_2}\bar{\xi_{k_2'}}\bigr]=\delta_0(k_1-k_1')\delta_0(k_2-k_2')+\delta_0(k_1+k_2)\delta_0(k_1'+k_2')+\delta_0(k_1-k_2')\delta_0(k_1'-k_2),
\end{equation*}
we get
\begin{align*}
\IE\bigl[\bigl\lvert\nabla Y_\eps(x)\bigr\rvert^2\bigr]&=\sum_{\substack{k\neq k'\\k,k'\neq 0}}\!\!\theta\bigl(\eps\abs{k}\bigr)^2\theta\bigl(\eps\abs{k'}\bigr)^2\frac{(k\cdot k')^2}{\abs{k}^4\abs{k'}^4\abs{k\mk -\mk k'}^2}+\sum_{\substack{k_1\neq k_2\\k_1,k_2\neq 0}}\!\!\theta\bigl(\eps\abs{k_1}\bigr)^2\theta\bigl(\eps\abs{k_2}\bigr)^2\frac{(k_1\cdot k_2)^2}{\abs{k_1}^4\abs{k_2}^4\abs{k_1\mk -\mk k_2}^2}\\
&=2\sum_{\substack{k\neq k'\\k,k'\neq 0}}\theta\bigl(\eps\abs{k}\bigr)^2\theta\bigl(\eps\abs{k'}\bigr)^2\frac{(k\cdot k')^2}{\abs{k}^4\abs{k'}^4\abs{k-k'}^2}
\end{align*}
which diverges logarithmically in $\eps$. As the process is invariant in law by translation again, the expectation is a constant function hence
\begin{equation*}
\Delta_n\Bigl(\abs{\nabla Y_\eps}^2-\IE\bigl[\abs{\nabla Y_\eps}^2\bigr]\Bigr)(x)=\sum_{m\ge1}K_n(m)\widehat{\abs{\nabla Y_\eps}^2}(m)e^{im\cdot x}
\end{equation*}
for $n\ge1$. We obtain that $\Bigl\lvert\Delta_n\Bigl(\abs{\nabla Y_\eps}^2-\IE\bigl[\abs{\nabla Y_\eps}^2\bigr]\Bigr)\Bigr\rvert^2(x)$ is of order
\begin{multline*}
2^{3n}\bigl\lvert\widehat{\abs{\nabla Y_\eps}^2}(2^n)\bigr\rvert^2\\
=2^{3n}\sum_{\substack{k_1 - k_1' + k_2 - k_2'=2^n\\ \abs{k_i},\abs{k_i'}\le\eps^{-1}}}\sum_{\substack{\ell_1 - \ell_1' +\ell_2 - \ell_2'=2^n\\ \abs{\ell_i},\abs{\ell_i'}\le\eps^{-1}}}\frac{k_1\cdot k_1'}{\abs{k_1}^2\abs{k_1'}^2}\frac{k_2\cdot k_2'}{\abs{k_2}^2\abs{k_2'}^2}\frac{(k_1 -  k_1')\cdot(k_2' -  k_2)}{\abs{k_1 -  k_1'}^2\abs{k_2 -  k_2'}^2}\xi_{k_1}\bar{\xi_{k_1'}}\xi_{k_2'}\bar{\xi_{k_2}}\\
\frac{\ell_1\cdot\ell_1'}{\abs{\ell_1}^2\abs{\ell_1'}^2}\frac{\ell_2\cdot\ell_2'}{\abs{\ell_2}^2\abs{\ell_2'}^2}\frac{(\ell_1-\ell_1')\cdot(\ell_2'-\ell_2)}{\abs{\ell_1-\ell_1'}^2\abs{\ell_2-\ell_2'}^2}\bar{\xi_{\ell_1}}\xi_{\ell_1'}\bar{\xi_{\ell_2'}}\xi_{\ell_2}
\end{multline*}
using that $K_n$ is supported in an annulus of $\IZ^3$ of radius $2^n$. Taking the expectation gives a factor
\begin{equation*}
\IE\bigl[\xi_{k_1}\bar{\xi_{k_1'}}\xi_{k_2'}\bar{\xi_{k_2}}\bar{\xi_{\ell_1}}\xi_{\ell_1'}\bar{\xi_{\ell_2'}}\xi_{\ell_2}\bigr]
\end{equation*}
which is the sum of a product of delta correlation functions over all possible pairings, that is 28 elements. A number of terms vanish due to the spectral localization as in the previous proof on $\IT^2$, for example the paring $\delta_0(k_1-k_1')\delta_0(k_2-k_2')\delta_0(\ell_1-\ell_1')\delta_0(\ell_2-\ell_2')$. An example of non-vanishing paring is
\begin{equation*}
\delta_0(k_1-k_2)\delta_0(k_1'-k_2')\delta_0(\ell_1-\ell_2)\delta_0(\ell_1'-\ell_2')
\end{equation*}
which gives the sum
\begin{multline*}
\sum_{\substack{2k\mk - \mk 2k'=2^n\\ \abs{k},\abs{k'}\le\eps^{-1}}}\,\sum_{\substack{2\ell\mk - \mk 2\ell'=2^n\\ \abs{\ell},\abs{\ell'}\le\eps^{-1}}}\frac{(k\cdot k')^2}{\abs{k}^4\abs{k'}^4}\frac{1}{\abs{k\Mk - \Mk k'}^2}\frac{(\ell\cdot\ell')^2}{\abs{\ell}^4\abs{\ell'}^4}\frac{1}{\abs{\ell\Mk - \Mk \ell'}^2}\\
\begin{aligned}
&=2^{-2(n-1)}\sum_{\substack{2k\mk - \mk 2k'=2^n\\ \abs{k},\abs{k'}\le\eps^{-1}}}\,\sum_{\substack{2\ell\mk - \mk 2\ell'=2^n\\ \abs{\ell},\abs{\ell'}\le\eps^{-1}}}\frac{(k\cdot k')^2}{\abs{k}^4\abs{k'}^4}\frac{(\ell\cdot\ell')^2}{\abs{\ell}^4\abs{\ell'}^4}\\
&=2^{-2(n-1)}\left(\sum_{\substack{2k-2k'=2^n\\ \abs{k},\abs{k'}\le\eps^{-1}}}\frac{(k\cdot k')^2}{\abs{k}^4\abs{k'}^4}\right)^2
\le2^{-2(n-1)}\left(\sum_{\substack{2k-2k'=2^n\\ \abs{k},\abs{k'}\le\eps^{-1}}}\frac{1}{\abs{k}^2\abs{k'}^2}\right)^2\\
&\le2^{-2(n-1)}\left(\sum_{\abs{k}\le\eps^{-1}}\frac{1}{\abs{k}^2\abs{k-2^{n-1}}^2}\right)^2
\le C2^{-2(n-1)}2^{-(n-1)(1-\kappa)}\left(\sum_{\abs{k}\le\eps^{-1}}\frac{1}{\abs{k}^{3+\kappa}}\right)^2
\end{aligned}
\end{multline*}
where the sum over $k\in\IZ^3\backslash\{0\}$ is convergent. Since there are $2^{3n}$ terms in the sum because of the support of $K_n$, this will indeed gives a bound of the order
\begin{equation*}
\IE\Bigl\lvert\Delta_n\Bigl(\abs{\nabla Y_\eps}^2-\IE\bigl[\abs{\nabla Y_\eps}^2\bigr]\Bigr)\Bigr\rvert^2\le C2^{\kappa n}.
\end{equation*}
The large number of terms appearing here is one of the problems of singular SPDEs and it is the main motivation for the introduction of general algebraic structures based on Hopf algebra of trees and diagrams in regularity structures, see Hairer's seminal work~\cite{Hai14}. A control of each term yields the bound
\begin{equation*}
\IE\Bigl\lvert\Delta_n\Bigl(\abs{\nabla Y_\eps}^2-\IE\bigl[\abs{\nabla Y_\eps}^2\bigr]\Bigr)\Bigr\rvert^p\le C_p2^{\kappa np}
\end{equation*}
using Gaussian hypercontractivity again and so the sequence is bounded in $$\CB_{p,p}^{-\kappa}(\IT^3)\longhookrightarrow\CC^{-\kappa-{\frac{3}{p}}}(\IT^3).$$ Similar computations give that the sequence is Cauchy and completes the proof. For $\nabla X\diamond\nabla Y$, the computations are similar but simpler since this is only a trilinear functional of the noise while $\abs{\nabla Y}^{2\diamond}$ is a $4$-linear functional of the noise. In this case, the renormalization constant is zero because of algebraic cancellations, see for example~\cite[Section~9]{GubinelliPerkowski17} for similar computations.
\end{proof}

We define the three dimensional enhanced noise
\begin{equation*}
\Xi=\bigl(\xi,\abs{\nabla X}^{2\diamond},\abs{\nabla Y}^{2\diamond},\nabla X\diamond\nabla Y\bigr)
\end{equation*}
which belongs to
\begin{equation*}
\CX^\kappa(\IT^3)=\CC^{-\frac{3}{2}-\kappa}(\IT^3)\times\CC^{-1-2\kappa}(\IT^3)\times\CC^{-4\kappa}(\IT^3)\times\CC^{-\frac{1}{2}-3\kappa}
\end{equation*}
for any $\kappa>0$. We also have that
\begin{equation*}
\Xi_\eps=\bigl(\xi_\eps,\abs{\nabla X_\eps}^2-(2\pi)^{-2}\eps^{-1},\abs{\nabla Y_\eps}^2-(2\pi)^{-2}\log(\eps),\nabla X_\eps\cdot\nabla Y_\eps\bigr)
\end{equation*}
converges to $\Xi$ in $\CX^\kappa(\IT^3)$ for any $\kappa>0$.

\section{Construction in two dimensions} \label{Section2D}

It is tempting to define the form of the Anderson operator by
\begin{equation*}
{\mathfrak a}(u_1,u_2)=\int_{\IT^2}\nabla u_1(x)\cdot\nabla u_2(x)\drm x+\int_{\IT^2}u_1(x)u_2(x)\xi(\drm x)
\end{equation*}
for any $u_1,u_2\in C^\infty(\IT^2)$. However, this is not a natural object since this form is not closable as shown by the recent progress on singular stochastic operators, which can be guessed from the fact that for $u\in\CH^1$ the form domain of $\Delta$, the product $u\xi$ is ill-defined. For $\xi_\eps=\xi*\rho_\eps$ a regularization of the noise, consider the regularized form
\begin{equation*}
{\mathfrak a}_\eps(u_1,u_2)=\int_{\IT^2}\nabla u_1(x)\cdot\nabla u_2(x)\drm x+\int_{\IT^2}u_1(x)u_2(x)\bigl(\xi_\eps(x)-c_\eps\bigr)\drm x
\end{equation*}
with $c_\eps$ the logarithmic diverging constant defined in the previous section. For any fixed $\eps>0$, ${\mathfrak a}_\eps$ is a closed symmetric form with domain $\CH^1$ and we construct a form ${\mathfrak a}$ such that ${\mathfrak a}_\eps$ converges to ${\mathfrak a}$ as $\eps$ goes to $0$. With $X$ the random field constructed in the previous section, we consider the new variable $u=e^Xv$ and define
\begin{equation*}
Hu\coloneqq -e^X\Delta v-2e^X\nabla X\cdot\nabla v-e^X\abs{\nabla X}^{2\diamond}v+\xi_0e^Xv
\end{equation*}
for $v\in\CC^\infty$. Since $X\in\CC^{1-\kappa}$ and $\abs{\nabla X}^{2\diamond}\in\CC^{-2\kappa}$, $Hu$ is well-defined as a distribution. The associated form is given by
\begin{align*}
\mathfrak{a}(u_1,u_2)&=\langle Hu_1,u_2\rangle\\
&= \bigl\langle He^Xv_1,e^Xv_2\bigr\rangle\\
&=-\bigl\langle\Delta v_1,v_2e^{2X}\bigr\rangle-2\bigl\langle \nabla X\cdot\nabla v_1,v_2e^{2X}\bigr\rangle-\bigl\langle\abs{\nabla X}^{2\diamond}v_1,v_2e^{2X}\bigr\rangle+\xi_0\bigl\langle v_1,v_2e^{2X}\bigr\rangle\\
&=\int_{\IT^2}\nabla v_1(x)\cdot\nabla v_2(x)e^{2X(x)}\drm x-\bigl\langle\abs{\nabla X}^{2\diamond}v_1,v_2e^{2X}\bigr\rangle+\xi_0\int_{\IT^2}v_1(x)v_2(x)e^{2X(x)}\drm x
\end{align*}
which is well-defined for $v_1,v_2\in\CH^1$ since
\begin{align*}
\Bigl\lvert\bigl\langle\abs{\nabla X}^{2\diamond}e^{2X},v_1v_2\bigr\rangle\Bigr\rvert&\le\bigl\lVert \abs{\nabla X}^{2\diamond}e^{2X}\bigr\rVert_{\CC^{-\kappa}}\norm{v_1v_2}_{\CB_{1,1}^\kappa}\\
&\le\bigl\lVert \abs{\nabla X}^{2\diamond}\bigr\rVert_{\CC^{-\kappa}}\norm{e^{2X}}_{\CC^{2\kappa}}\norm{v_1}_{\CH^{2\kappa}}\norm{v_2}_{\CH^{2\kappa}}\\
&\le\bigl\lVert \abs{\nabla X}^{2\diamond}\bigr\rVert_{\CC^{-\kappa}}\norm{e^{2X}}_{\CC^{1-\kappa}}\norm{v_1}_{\CH^1}\norm{v_2}_{\CH^1}
\end{align*}
for $\kappa>0$ small enough using Propositions~\ref{PropDuality} and~\ref{PropProduct}.

\begin{definition}
The Anderson form is defined by
\begin{equation*}
{\mathfrak a}(u_1,u_2)\coloneqq \langle \nabla v_1,\nabla v_2\rangle_{L^2(\IT^2,e^{2X}\drm x)}-\bigl\langle\abs{\nabla X}^{2\diamond},v_1v_2e^{2X}\bigr\rangle+\xi_0\langle v_1,v_2\rangle_{L^2(\IT^2,e^{2X}\drm x)}
\end{equation*}
where $v_i=e^{-X}u_i$ with domain $\CD({\mathfrak a})\coloneqq e^X\CH^1$ equipped with the norm
\begin{equation*}
\norm{u}_{\mathfrak a}^2\coloneqq \norm{u}_{L^2}^2+\norm{e^{-X}u}_{\CH^1}^2.
\end{equation*}
\end{definition}

Since $e^X\in\CC^{1-\kappa}$ for any $\kappa>0$, the domain $\CD(\mathfrak{a})$ is dense in $\CH^{1-\kappa}$ thus in $L^2$. The following proposition states that this densely defined form is continuous and bounded from below.
\begin{proposition}
There exists a random constant $C>0$ such that
\begin{equation*}
\bigl\lvert\mathfrak{a}(u_1,u_2)\bigr\rvert\le C\norm{u_1}_{\mathfrak a} \norm{u_2}_{\mathfrak a}
\end{equation*}
for $u_1,u_2\in\CD(\mathfrak{a})$. The form ${\mathfrak a}$ is quasi-coercive, i.e., there exist random constants $ \delta, C'>0$ such that
\begin{equation*}
{\mathfrak a}(u,u) + C' \norm{u}_{L^2}^2 \ge \delta \norm{u}_{\mathfrak a}^2
\end{equation*}
for all $u=e^Xv\in\CD({\mathfrak a})$.
\end{proposition}

\begin{proof}
The continuity follows directly from
\begin{equation*}
\Bigl\lvert\bigl\langle\abs{\nabla X}^{2\diamond}e^{2X},v_1v_2\bigr\rangle\Bigr\rvert\le\bigl\lVert \abs{\nabla X}^{2\diamond}\bigr\rVert_{\CC^{-\kappa}}\norm{e^{2X}}_{\CC^{1-\kappa}}\norm{v_1}_{\CH^1}\norm{v_2}_{\CH^1}.
\end{equation*}
Now we prove the second statement. Set $u = e^X v$ with $v \in \CH^1$. We have for any $\kappa >0$
\begin{align*}
\mathfrak{a}(u,u) - \xi_0\int_{\IT^2}\bigl\lvert v(x)\bigr\rvert^2e^{2X(x)}\drm x &= \int_{\IT^2} \bigl\lvert\nabla v(x)\bigr\rvert^2(x)e^{2X(x)}\drm x -\bigl\langle\abs{\nabla X}^{2\diamond}v,ve^{2X}\bigr\rangle\\
&\ge e^{-\norm{X}_{L^\infty}} \int_{\IT^2} \bigl\lvert\nabla v(x)\bigr\rvert^2(x)\drm x -\bigl\lVert \abs{\nabla X}^{2\diamond}\bigr\rVert_{\CC^{-\kappa}}\norm{e^{2X}}_{\CC^{2\kappa}}\norm{v}_{\CH^{2\kappa}}^2.
\end{align*}
For small $\kappa > 0$ we use the standard interpolation inequality, which is valid for every $\varepsilon > 0$,
\begin{equation*}
\norm{v}_{\CH^{2\kappa}} \le \varepsilon \norm{v}_{\CH^{1}} + c_\varepsilon \norm{v}_{L^2}
\end{equation*}
for some $c_\varepsilon > 0$. We choose $\varepsilon$ small enough and insert this inequality in the previous estimates to obtain the statement.
\end{proof}

A consequence of the previous proposition is that the norms $\norm{\,\cdot\,}_{\CD(\mathfrak{a})}$ and $\norm{e^{-X}\,\cdot\,}_{\CH^1}$ are equivalent.

We now prove that the form is closed.

\begin{proposition}
The form $\mathfrak{a}$ is closed, that is $\bigl(\CD(\mathfrak{a}),\norm{\,\cdot\,}_\mathfrak{a}\bigr)$ is a complete space.
\end{proposition}

\begin{proof}
Let $(u_n)_{n\ge0}\subset\CD(\mathfrak{a})$ be a Cauchy sequence. Then $(e^{-X}u_n)_{n\ge0}$ is a Cauchy sequence in $\CH^1$ thus converges to a limit $v\in\CH^1$ while $(u_n)_{n\ge0}$ is a Cauchy sequence in $L^2$ thus converges to $u\in L^2$. We have
\begin{align*}
\norm{u-e^Xv}_{L^2}&\le\norm{u-u_n}_{L^2}+\norm{u_n-e^Xv}_{L^2}\\
&\le\norm{u-u_n}_{L^2}+\norm{e^X}_{L^\infty}\norm{e^{-X}u_n-v}_{L^2}
\end{align*}
Letting $n \to \infty$ yields $u=e^Xv \in \CD(\mathfrak{a})$ and this completes the proof.
\end{proof}

Finally, we prove that $\mathfrak{a}$ is the limit in some sense of the renormalized forms $\mathfrak{a}_\eps$.

\begin{proposition}
For any $\kappa>0$, there exists a constant $C>0$ such that
\begin{equation*}
\bigl\lvert\mathfrak{a}(u_1,u_2)-\mathfrak{a}_\eps(u_1^\eps,u_2^\eps)\bigr\rvert\le C\norm{\Xi-\Xi_\eps}_{\CX^\kappa(\IT^2)}\norm{v_1}_{\CH^1}\norm{v_2}_{\CH^1}
\end{equation*}
with $u_i^\eps=e^{X_\eps}v_i$ for $\eps\ge0$.
\end{proposition}

\begin{proof}
Let $v_1,v_2\in\CH^1$ and consider $u_i^\eps=e^{X_\eps}v_i$ for $\eps\ge0$, i.e., $u_i^\eps\in\CH^1$ the form domain of $a_\eps$ for any $\eps>0$ while $u_i\in e^X\CH^1$ for $\eps=0$. We have
\begin{equation*}
\mathfrak{a}_\eps(u_1^\eps,u_2^\eps)=\langle \nabla v_1,\nabla v_2\rangle_{L^2(\IT^2,e^{2X}\drm x)}-\bigl\langle\abs{\nabla X_\eps}^2-c_\eps,v_1v_2e^{2X_\eps}\bigr\rangle+\langle\xi_\eps,1\rangle\langle v_1,v_2\rangle_{L^2(\IT^2,e^{2X_\eps}\drm x)}
\end{equation*}
and hence
\begin{align*}
\bigl\lvert\mathfrak{a}(u_1,u_2)-\mathfrak{a}_\eps(u_1^\eps,u_2^\eps)\bigr\rvert&\le\Bigl\lvert\bigl\langle\abs{\nabla X_\eps}^2-c_\eps-\abs{\nabla X}^{2\diamond},v_1v_2\bigr\rangle\Bigr\rvert+\Bigl\lvert\bigl\langle\langle\xi_\eps,1\rangle e^{X_\eps}-\xi_0e^X,v_1v_2\bigr\rangle\Bigr\rvert\\
&\le C\norm{\Xi-\Xi_\eps}_{\CX^\kappa}\norm{v_1}_{\CH^1}\norm{v_2}_{\CH^1}
\end{align*}
for any $\kappa>0$ and the proof is complete.
\end{proof}

\section{Construction in three dimensions} \label{Section3D}

In three dimensions, the expression
\begin{equation*}
\langle \nabla v_1,\nabla v_2\rangle_{L^2(\IT^3,e^{2X}\drm x)}-\bigl\langle\abs{\nabla X}^{2\diamond},v_1v_2e^{2X}\bigr\rangle+\xi_0\bigl\langle v_1,v_2e^{2X}\bigr\rangle
\end{equation*}
does not make sense anymore for $v_1,v_2\in\CH^1$ since $\abs{\nabla X}^{2\diamond}$ belongs to $\CC^{-1-\kappa}$ for any $\kappa>0$. In this case, one makes the change of variable $u=e^{X+Y}v$ with $Y$ the solution to
\begin{equation*}
\Delta Y=\abs{\nabla X}^{2\diamond}-\bigl\langle\abs{\nabla X}^{2\diamond},1\bigr\rangle
\end{equation*}
which belongs to $\CC^{1-\kappa}$. We have
\begin{equation*}
Hu=-e^{X+Y}\Delta v-2e^{X+Y}(\nabla X+\nabla Y)\cdot\nabla v-\Bigl(\abs{\nabla Y}^{2\diamond}+2\nabla X\diamond\nabla Y-\bigl\langle \abs{\nabla X}^{2\diamond},1\bigr\rangle-\xi_0\Bigr)e^{X+Y}v
\end{equation*}
hence
\begin{equation*}
\begin{split}
\mathfrak{a}(u_1,u_2)&=\langle Hu_1,u_2\rangle\\
&=\bigl\langle He^{X+Y}v_1,e^{X+Y}v_2\bigr\rangle\\
&=-\bigl\langle \Delta v_1,v_2e^{2X+2Y}\bigr\rangle-2\bigl\langle \nabla(X+Y)\cdot\nabla v_1,v_2e^{2X+2Y}\bigr\rangle-\bigl\langle\abs{\nabla Y}^{2\diamond},v_1v_2e^{2X+2Y}\bigr\rangle\\
&\qquad\qquad-2\bigl\langle\nabla X\diamond\nabla Y,v_1v_2e^{2X+2Y}\bigr\rangle+\bigl\langle\abs{\nabla X}^{2\diamond},1\bigr\rangle+\bigl\langle\xi_0,v_1v_2e^{2X+2Y}\bigr\rangle\\
&=\int_{\IT^3}\nabla v_1(x)\cdot\nabla v_2(x)e^{2X(x)+2Y(x)}\drm x-\bigl\langle\abs{\nabla Y}^{2\diamond}+2\nabla X\diamond\nabla Y,v_1v_2e^{2X+2Y}\bigr\rangle\\
&\qquad\qquad+\bigl(\bigl\langle\abs{\nabla X}^{2\diamond},1\bigr\rangle+\xi_0\bigr)\int_{\IT^3}v_1(x)v_2(x)e^{2X(x)+2Y(x)}\drm x
\end{split}
\end{equation*}
which is well-defined for $v_1,v_2\in\CH^1$ since $\abs{\nabla Y}^{2\diamond}\in\CC^{-\kappa}$ and $\nabla X\diamond\nabla Y\in\CC^{-\frac{1}{2}-\kappa}$ for any $\kappa>0$.

\begin{definition}
The Anderson form is defined by
\begin{multline*}
\mathfrak{a}(u_1,u_2)\coloneqq \langle \nabla v_1,\nabla v_2\rangle_{L^2(\IT^3,e^{2X+2Y}\drm x)}-\bigl\langle\abs{\nabla Y}^{2\diamond}+2\nabla X\diamond\nabla Y,v_1v_2e^{2X+2Y}\bigr\rangle\\
+\bigl(\bigl\langle\abs{\nabla X}^{2\diamond},1\bigr\rangle+\xi_0\bigr)\langle v_1,v_2\rangle_{L^2(\IT^3,e^{2X+2Y}\drm x)}
\end{multline*}
where $v_i=e^{-X}u_i$ with domain $\CD(\mathfrak{a})\coloneqq e^{X+Y}\CH^1$ equipped with the norm
\begin{equation*}
\norm{u}_\mathfrak{a}^2\coloneqq \norm{u}_{L^2}^2+\norm{e^{-(X+Y)}u}_{\CH^1}^2.
\end{equation*}
\end{definition}

Since $e^{X+Y}\in\CC^{\frac{1}{2}-\kappa}$ for any $\kappa>0$, the domain $\CD(\mathfrak{a})$ is dense $\CH^{\frac{1}{2}-\kappa}$ thus in $L^2$. The following proposition states that this densely defined form is continuous and bounded from below. The proofs are obtained following the same path as in two dimensions.

\begin{proposition}
There exists a random constant $C>0$ such that
\begin{equation*}
\bigl\lvert\mathfrak{a}(u_1,u_2)\bigr\rvert\le C\norm{u_1}_\mathfrak{a}\norm{u_2}_\mathfrak{a}
\end{equation*}
for $u_1,u_2\in\CD(\mathfrak{a})$. There exists random constants $\delta, C'>0$ such that
\begin{equation*}
{\mathfrak a}(u,u) + C' \norm{u}_{L^2}^2 \ge \delta \norm{u}_{\mathfrak a}^2
\end{equation*}
for all $u=e^{X+Y}v\in\CD(\mathfrak{a})$.

Moreover, the form $\mathfrak{a}$ is closed, that is $\bigl(\CD(\mathfrak{a}),\norm{\,\cdot\,}_\mathfrak{a}\bigr)$ is a complete space, and we have the following convergence. For any $\kappa>0$, there exists a constant $C>0$ such that
\begin{equation*}
\bigl\lvert\mathfrak{a}(u_1,u_2)- \mathfrak{a}_\eps(u_1^\eps,u_2^\eps)\bigr\rvert\le C\norm{\Xi-\Xi_\eps}_{\CX^\kappa(\IT^3)}\norm{v_1}_{\CH^1}\norm{v_2}_{\CH^1}
\end{equation*}
with $u_i^\eps=e^{X_\eps+Y_\eps}v_i$ for $\eps\ge0$.
\end{proposition}

\section{Positivity and spectral gap} \label{SectionSpectralGap}

The construction of the form $\mathfrak{a}$ is the same in two and three dimensions. It is densely defined, symmetric bounded from below, continuous and closed. Its associated operator $H$ has domain
\begin{equation*}
\CD(H)=\Bigl\{u\in L^2(\IT^d)\ ;\ \exists v\in L^2(\IT^d),\,\forall\varphi\in\CD(\mathfrak{a}), \ \mathfrak{a}(u,\varphi)=\langle v,\varphi\rangle\Bigr\}.
\end{equation*}
This is well-known in the study of spectral theory, see for example~\cite[Section~1.2.3]{Ouhabaz05}. The idea is that for $u\in\CD(H)$, the bilinear form
\begin{equation*}
\varphi\in\CD(\mathfrak{a})\mapsto\mathfrak{a}(u,\varphi)
\end{equation*}
can be represented as a unique element $v\in L^2(\IT^d)$ and has to be interpreted as
\begin{equation*}
\varphi\mapsto\langle v,\varphi\rangle=\langle H u,\varphi\rangle
\end{equation*}
with $H$ the operator associated to the form $\mathfrak{a}$. The density of the form domain is crucial and one can transfer most property of the form to the operator, the domain of the operator being dense in the form domain. In particular, the operator $H$ is self-adjoint, densely defined and bounded from below. Since $\CD(\mathfrak{a})$ is embedded into a Sobolev space of positive regularity, it is compactly embedded in $L^2(\IT^d)$. Therefore, $H$ has discrete spectrum
\begin{equation*}
\lambda_1\le\lambda_2\le\dots
\end{equation*}
and there exists an orthonormal basis of $L^2(\IT^d)$ which is given by eigenfunctions of $H$. An important information is the existence of a spectral gap with a positive ground state. This is already known (see for example~\cite{BailleulDangMouzard22}) and it is a key to prove two-sided Gaussian bounds for the corresponding heat kernel of $H$. By the classical Krein--Rutman theorem, the general idea to get a spectral gap with a positive ground state is to prove that the semigroup $e^{-tH}$ is positive and irreducible. This means that for any non-negative (and nontrivial) $f \in L^2(\IT^d)$, we have at any time $t > 0$, $e^{-tH} f > 0$ a.e.\ on $\IT^d$. The irreducibility is sometimes called \emph{strict positivity} or \emph{positivity improving}. Unlike~\cite{BailleulDangMouzard22} which relies on quantitative study of the linear Parabolic Anderson Model equation and an approximation argument, we can obtain positivity and irreducibility readily from the form. These two properties are indeed characterized in terms of the form. See Theorems~$2.6$ and $2.10$ in~\cite{Ouhabaz05}. Thus, we provide a pedestrian approach to the existence of a spectral gap even in three dimensions which usually relies on involved computations with expansion of order $5$ using regularity structures or paracontrolled calculus.

\begin{theorem}
The semigroup $e^{-tH}$ is irreducible. In particular, the first eigenvalue is simple, that is $\lambda_1<\lambda_2$ and there exists a positive ground state $\Psi\in\CD(H)$.
\end{theorem}

\begin{proof}
Both positivity and irreducibility are not changed under multiplication by $e^{X}$ or $e^{X+Y}$ and so we use the form $\mathfrak{a}$ constructed in the previous sections.

Let $u \in D(\mathfrak{a})$ and $v \in \CH^1$ such that $u = e^{X} v$ if $d=2$ and $u = e^{X+Y} v$ if $d=3$. Then clearly, $u^+ = e^{X} v^+$ (or $e^{X+Y} v^+$) and $u^- = e^X v^-$ (or $e^{X+Y} v^-$). Since $v^+, v^- \in \CH^1$, we have $u^+, u^- \in D(\mathfrak{a})$. In addition, it is obviously seen from the definition of the Anderson form that $ \mathfrak{a}(u^+, u^-) = 0$. By~\cite[Theorem~2.6]{Ouhabaz05} we conclude that $(e^{-tH})_{t\ge 0}$ is a positive semigroup.

Now we prove irreducibility. We apply~\cite[Theorem~2.10]{Ouhabaz05}. Since $H$ is a local operator, it is enough to prove that if $F\subset\IT^d$ is such that
\begin{equation*}
\forall u\in D(\mathfrak{a}),\quad \IDC_F u\in D(\mathfrak{a}),
\end{equation*}
then either $\abs{F}=0$ or $\abs{\IT^d\backslash F}=0$. Clearly, $ \IDC_F u = e^{X} \IDC_F v $ if $d=2$ and $ \IDC_F u = e^{X +Y} \IDC_F v $ if $d= 3$. This implies
\begin{equation*}
\forall v\in\CH^1,\quad\IDC_Fv\in\CH^1.
\end{equation*}
Theorem~2.10 from~\cite{Ouhabaz05} applied to the Laplacian, whose form domain is $\CH^1$, gives that $\abs{F}=0$ or $\abs{\IT^d\backslash F}=0$. This proves irreducibility.

The rest of the theorem is classical and it is a direct consequence of the Krein--Rutman theorem.
\end{proof}

\begin{remark}
To have a more precise characterization of the domain, one would need to understand the condition $Hu\in L^2(\IT^d)$ for $u\in L^2(\IT^d)$. For example in the case $d=2$, smooth functions are not in the domain of the operator since
\begin{equation*}
e^{-X}He^Xv=-\Delta v-2\nabla X\cdot\nabla v-\abs{\nabla X}^{2\diamond}v
\end{equation*}
is not an element of $L^2(\IT^2)$ for smooth $v$ since $\nabla X,\abs{\nabla X}^{2\diamond}\in\CC^{-\kappa}(\IT^2)$ for any $\kappa>0$. The main idea of the construction of such singular stochastic operator is to consider rough functions depending on the noise such that
\begin{equation*}
-\Delta u+u\xi\in L^2(\IT^2)
\end{equation*}
hence $\Delta u$ needs to have the same regularity of the noise, that is $u\in\CC^{1-\kappa}(\IT^2)$. Using regularity structures or paracontrolled calculus gives a local description of such function to ensure not only to define the operator with a renormalization procedure but also to give explicit functions in the domain with a prescribed local behavior depending on the noise. In the framework of~\cite{Mouzard}, the form domain is described with a first order paracontrolled expansion
\begin{equation*}
\CD(\mathfrak{a})=\Bigl\{u\in L^2(\IT^2)\ ;\ u-\P_uX\in\CH^1(\IT^2)\Bigr\}
\end{equation*}
where $\P$ is a paraproduct, which is equivalent to our formulation. However, the operator domain is obtained with a second order paracontrolled expansion
\begin{equation*}
\CD(H)=\Bigl\{u\in L^2(\IT^2)\ ;\ u-\P_uX-\P_uX_2\in\CH^2(\IT^2)\Bigr\}
\end{equation*}
with $X_2$ another suitable function depending on $\xi$ and $X$ which does not involve any renormalization procedure.
\end{remark}

\section*{Declaration of interests}

The authors do not work for, advise, own shares in, or receive funds from any organization that could benefit from this article, and have declared no affiliations other than their research organizations.

\printbibliography
\end{document}